Author's response to reviews

Title: Efficacy of a pressure-sensing mattress cover system on reducing interface pressure: study protocol for a randomized controlled trial

Authors:

Holly Wong (hsywong@ucalgary.ca)
Jaime Kaufman (jbjarnso@ucalgary.ca)
Barry Baylis (baylis@ucalgary.ca)
John M Conly (john.conly@albertahealthservices.ca)
David B Hogan (dhogan@ucalgary.ca)
Henry T Stelfox (tstelfox@ucalgary.ca)
Danielle A Southern (dasouthe@ucalgary.ca)
William A Ghali (wghali@ucalgary.ca)
Chester H Ho (Chester.Ho@albertahealthservices.ca)

Version: 4  Date: 8 September 2015

Author's response to reviews:

Holly Wong
W21C Research and Innovation Centre
GD01 TRW Building
3280 Hospital Drive, NW
Calgary, AB, Canada T2N-4Z6

Enclosure

August 7, 2015

Dr. Doug Altman
Dr. Curt Furber
Dr. Jeremy Grimshaw
Editors-in-Chief
Trials

Re: Resubmission of MS: 1909557830162465 - Efficacy of a pressure-sensing mattress cover system on reducing interface pressure: Study protocol for a parallel two-group randomized trial

Dear Drs. Altman, Furberg and Grimshaw,

Thank you for the opportunity to address the formatting requests. In this submission we have:
- Changed the title format
- Stated the funding status in the Acknowledgements section
- Included information on our additional files after the Reference section

Thank you once again for reviewing our manuscript and allowing us to revise it for acceptance for publication in Trials.

Kind regards,
Holly Wong (on behalf of our team)